\begin{thebibliography}{1}

\bibitem{author:06}
First Author and Second Author.
\newblock Some publication title.
\newblock In Editor and Editor \cite{conference:06}, pages 330--331.

\bibitem{WebCamwhitePaper:dboulig}
Douglas Boulig.
\newblock Windows ce webcam driver white paper.

\bibitem{ProgrammingWinCE2ndEd:dboulig}
Douglas Boulig.
\newblock {\em Programming Microsoft Windows CE}.
\newblock Microsoft Press, second edition, 2001.

\bibitem{ProgrammingWinCE3rdEd:dboulig}
Douglas Boulig.
\newblock {\em Programming Microsoft Windows CE .NET}, volume 1224 pages.
\newblock Microsoft Press, third edition, 2003.

\bibitem{WindowsCEPratical:Beijing}
Zhang Dongquan, Tan Nanlin, Wang Xuemei, and Jiao Fengchuan.
\newblock {\em Windows CE Shi Yong Kai Fa Ji Shu}.
\newblock Publishing House of Electronics Industry, Beijing, 2006.

\bibitem{conference:06}
First Editor and Second Editor, editors.
\newblock {\em Proceedings of the Xth Conference on XYZ}, October 2006.

\bibitem{WindowsCEHandBook:TerenceAGoggin}
Terence~A. Goggin.
\newblock {\em Windows CE Developer's Handbook}.

\end{thebibliography}

%% Based on a TeXnicCenter-Template by Gyorgy SZEIDL.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%------------------------------------------------------------
%
\documentclass[12pt, a4paper]{article}%
%Options -- Point size:  10pt (default), 11pt, 12pt
%        -- Paper size:  letterpaper (default), a4paper, a5paper, b5paper
%                        legalpaper, executivepaper
%        -- Orientation  (portrait is the default)
%                        landscape
%        -- Print size:  oneside (default), twoside
%        -- Quality      final(default), draft
%        -- Title page   notitlepage, titlepage(default)
%        -- Columns      onecolumn(default), twocolumn
%        -- Equation numbering (equation numbers on the right is the default)
%                        leqno
%        -- Displayed equations (centered is the default)
%                        fleqn (equations start at the same distance from the right side)
%        -- Open bibliography style (closed is the default)
%                        openbib
% For instance the command
%           \documentclass[a4paper,12pt,leqno]{article}
% ensures that the paper size is a4, the fonts are typeset at the size 12p
% and the equation numbers are on the left side
%
\usepackage{amsmath}%
\usepackage{amsfonts}%
\usepackage{amssymb}%
\usepackage{graphicx}
%-------------------------------------------
\newtheorem{theorem}{Theorem}
\newtheorem{acknowledgement}[theorem]{Acknowledgement}
\newtheorem{algorithm}[theorem]{Algorithm}
\newtheorem{axiom}[theorem]{Axiom}
\newtheorem{case}[theorem]{Case}
\newtheorem{claim}[theorem]{Claim}
\newtheorem{conclusion}[theorem]{Conclusion}
\newtheorem{condition}[theorem]{Condition}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{criterion}[theorem]{Criterion}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{exercise}[theorem]{Exercise}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{notation}[theorem]{Notation}
\newtheorem{problem}[theorem]{Problem}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{solution}[theorem]{Solution}
\newtheorem{summary}[theorem]{Summary}
\newenvironment{proof}[1][Proof]{\textbf{#1.} }{\ \rule{0.5em}{0.5em}}

\setlength{\parindent}{0cm}


\begin{document}

\title{Master Thesis}
\author{Maoan Zhou}
\date{\today}


\maketitle

\newpage
\tableofcontents
\newpage

\begin{abstract}
abstract ...
\end{abstract}

\section{Introduction}

\section{Related work}
\subsection{Dynamic critiques by Apriori}
\subsection{Preference-based orgnization}
\subsection{Dataset}


\section{Human Intelligent Clustering}
\subsection{Human Intelligent Clustering's Interface}
\subsection{Human Intelligent Clustering's algorithm}

\section{Experiment result}

\section{Conclusion and future work}




This is ref~\cite{ProgrammingWinCE2ndEd:dboulig}

This is ref~\cite{ProgrammingWinCE3rdEd:dboulig}

This is ref~\cite{WebCamwhitePaper:dboulig}

This is ref~\cite{WindowsCEHandBook:TerenceAGoggin}

This is ref~\cite{WindowsCEPratical:Beijing}

This is ref~\cite{author:06} 

This is ref~\cite{conference:06}




\newpage
\bibliographystyle{plain}
\bibliography{my_ref}

\end{document}
